Author's response to reviews

Title: The interferon gamma gene polymorphism +874 A/T is associated with severe acute respiratory syndrome

Authors:

Wai Po Chong (h9820905@graduate.hku.hk)
WK Eddie Ip (WIP@partners.org)
Gloria Hoi Wan Tso (h98083405@hkusua.hku.hk)
Man Wai Ng (ivy_natsu@yahoo.com.hk)
Wilfred Hing Sang Wong (whswong@hkcc.hku.hk)
Helen Ka Wai Law (hkwlaw@hkcc.hku.hk)
Raymond WH Yung (rwhyung@ha.org.hk)
Eudora Y Chow (chowe@ha.org.hk)
KL Au (klau@ha.org.hk)
Eric YT Chan (eytchan@ha.org.hk)
Wilina Lim (wlim@pacific.net.hk)
JS Malik Peiris (malik@hkcc.hku.hk)
Yu Lung Lau (lauylung@hkcc.hku.hk)

Version: 3 Date: 2 May 2006

Author’s response to reviews:

Dr. Melissa Norton
Medical Editor-BMC Infectious diseases
BioMed Central Ltd,
Middlesex House,
34-42 Cleveland Street,
London W1T 4LB, UK

Dear Dr. Melissa Norton,

Re: The interferon gamma gene polymorphism +874 A/T is associated with severe acute respiratory syndrome

We have followed the requests to do the following formatting changes
1) Delete the running title at title page
2) Change all heading and sub-heading to sentence case
3) We have modified the competing interests sentence as suggested.
4) We have change "Author's contributions" to "Authors' contributions"

Yu-Lung Lau